

## **Author's response to reviews**

**Title:** SICTIN: Rapid footprinting of massively parallel sequencing data

### **Authors:**

Stefan Enroth ([stefan.enroth@lcb.uu.se](mailto:stefan.enroth@lcb.uu.se))  
Robin Andersson ([robin.andersson@lcb.uu.se](mailto:robin.andersson@lcb.uu.se))  
Claes Wadelius ([claes.wadelius@genpat.uu.se](mailto:claes.wadelius@genpat.uu.se))  
Jan Komorowski ([jan.komorowski@lcb.uu.se](mailto:jan.komorowski@lcb.uu.se))

**Version:** 3 **Date:** 12 August 2010

**Author's response to reviews:** see over

Changes made to the main text of the manuscript.

Major:

The references to the supplementary material are now written as e.g. “Additional file 1, Table S1”. A reference to “Additional file 2” has been inserted into “Availability and requirements”

Minor:

Full stop has been removed from title.

Track changes have been turned off.

Reference 9 has been updated with citation details.

Typography:

- i) Background, line 19: “basepair” change to “base pair”
- ii) Building the binaries, second paragraph, line 5-6: “an prologation” changed to “a prolongation”
- iii) Case study B, first paragraph, line 4: “align to many “ changed to “align to too many”.
- iv) Case study B, first paragraph, line 7: space added before reference 13 and 21.
- v) Figure 2, legend, line 2: “nuclsotides” changed to “nucleotides”
- vi) Figure 4, legend, line 6: “asterix” changed to “asterisk”
- vii) Additional files, Additional file 2, title: full stop removed.